Author's response to reviews

Title: Cystic Fibrosis and Renal Disease

Authors:

Baha A Al-Shawwa (balshaww@mcw.edu)
Aparna R Rao (arao@mcw.edu)

Version: 2 Date: 11 April 2007

Author's response to reviews:

Date: April 11th, 2007

The editor of Journal of Medical Case Reports,

We have revised our manuscript based on the recommendations and we are resubmitting it for your consideration.

Sincerely,

Baha Al-Shawwa, MD
Aparna Rao, MD